Author's response to reviews

Title: The need to modify physical activity messages to better speak to older African American women: a pilot study

Authors:

Emerson Sebastiao (esebast2@illinois.edu)
Wojtek Chodzko-Zajko (wojtek@illinois.edu)
Andiara Schwingel (andiara@illinois.edu)

Version: 6 Date: 16 September 2015

Author's response to reviews:

We have revised the manuscript in accordance with the concerns raised by the Editor. All the comments made by the Editor were followed. The revisions are highlighted in red in the revised submission.

A point-by-point response letter follows attached.